Author's response to reviews

Title: Antioxidant activity, total phenolic and total flavonoid contents of whole plant extracts Torilis leptophylla L.

Authors:

   Naima Saeed (naima_qau@hotmail.com)
   Muhammad R Khan (mrkhangau@yahoo.com)
   Maria Shabbir (comrademaria@gmail.com)

Version: 4 Date: 16 October 2012

Author's response to reviews:

Dear Editor,

manuscript is revised by one of our colleague who is more fluent in english. Now this manuscript appears to be acceptable for publication.

Dr Muhammad Rashid Khan
Associate Professor